Author’s response to reviews

Title: Cost-Effectiveness of Sofosbuvir-Based Treatments for Chronic Hepatitis C in the US

Authors:

Sai Zhang (swz5073@psu.edu)
Nathaniel D Bastian (nbastian@psu.edu)
Paul M Griffin (pgriffin@gatech.edu)

Version: 6 Date: 1 May 2015

Author’s response to reviews:

We received 4 editorial requests. They are listed below as well as our response to them.

1) Please move the tables to the end of the manuscript.
The tables were moved to the end, each table to a separate page.

2) Requesting to include a ‘Discussion’ section heading.
We now have both a Discussion and Conclusion section. This was accomplished by reorganizing the Conclusion section in the earlier version.

3) Requesting include a statement in the Authors’ Contributions confirming that all author read and approved the final version of the manuscript.
We have added this statement.

4) Please figure formatting according to the IforA.
We have reformatted the figure as requested.